\providecommand{\bysame}{\leavevmode\hbox to3em{\hrulefill}\thinspace}
\providecommand{\MR}{\relax\ifhmode\unskip\space\fi MR }
% \MRhref is called by the amsart/book/proc definition of \MR.
\providecommand{\MRhref}[2]{%
  \href{http://www.ams.org/mathscinet-getitem?mr=#1}{#2}
}
\providecommand{\href}[2]{#2}
\begin{thebibliography}{BCHM06}

\bibitem[BCHM06]{bchm}
Caucher Birkar, Paolo Cascini, Christopher~D. Hacon, and James McKernan,
  \emph{Existence of minimal models for varieties of log general type},
  http://www.citebase.org/abstract?id=oai:arXiv.org:math/0610203, 2006.

\bibitem[Che05]{che1}
Ivan Cheltsov, \emph{Birationally rigid {F}ano varieties}, Uspekhi Mat. Nauk
  \textbf{60} (2005), no.~5(365), 71--160. \MR{MR2195677 (2007d:14028)}

\bibitem[Che08]{che2}
\bysame, \emph{Fano varieties with many selfmaps}, Adv. Math. \textbf{217}
  (2008), no.~1, 97--124. \MR{MR2357324}

\bibitem[CM04]{co-me}
Alessio Corti and Massimiliano Mella, \emph{Birational geometry of terminal
  quartic 3-folds. {I}}, Amer. J. Math. \textbf{126} (2004), no.~4, 739--761.
  \MR{MR2075480 (2005d:14019)}

\bibitem[CR00]{co-re}
Alessio Corti and Miles Reid, \emph{Foreword}, Explicit birational geometry of
  3-folds, London Math. Soc. Lecture Note Ser., vol. 281, Cambridge Univ.
  Press, Cambridge, 2000, pp.~1--20.

\bibitem[Isk01]{isk}
V.~A. Iskovskikh, \emph{Birational rigidity of {F}ano hypersurfaces in the
  framework of {M}ori theory}, Uspekhi Mat. Nauk \textbf{56} (2001),
  no.~2(338), 3--86. \MR{MR1859707 (2002g:14017)}

\bibitem[KM92]{ko-mo}
J{\'a}nos Koll{\'a}r and Shigefumi Mori, \emph{Classification of
  three-dimensional flips}, J. Amer. Math. Soc. \textbf{5} (1992), no.~3,
  533--703. \MR{MR1149195 (93i:14015)}

\bibitem[Kol96]{rc-book}
J{\'a}nos Koll{\'a}r, \emph{Rational curves on algebraic varieties}, Ergebnisse
  der Mathematik und ihrer Grenzgebiete. 3. Folge. A Series of Modern Surveys
  in Mathematics, vol.~32, Springer-Verlag, Berlin, 1996. \MR{MR1440180
  (98c:14001)}

\bibitem[KSC04]{ksc}
J{\'a}nos Koll{\'a}r, Karen~E. Smith, and Alessio Corti, \emph{Rational and
  nearly rational varieties}, Cambridge Studies in Advanced Mathematics,
  vol.~92, Cambridge University Press, Cambridge, 2004. \MR{MR2062787
  (2005i:14063)}

\bibitem[Mel04]{mella}
Massimiliano Mella, \emph{Birational geometry of quartic 3-folds. {II}. {T}he
  importance of being {$\Bbb Q$}-factorial}, Math. Ann. \textbf{330} (2004),
  no.~1, 107--126. \MR{MR2091681 (2005h:14030)}

\bibitem[Ros56]{ros}
Maxwell Rosenlicht, \emph{Some basic theorems on algebraic groups}, Amer. J.
  Math. \textbf{78} (1956), 401--443. \MR{MR0082183 (18,514a)}

\bibitem[Shr08]{shr}
Constantin Shramov, \emph{Birational automorphisms of nodal quartic
  threefolds}, http://www.citebase.org/abstract?id=oai:arXiv.org:0803.4348,
  2008.

\end{thebibliography}

\documentclass{amsart}


%\documentstyle[amsmath,amssymb,amscd,graphicx,psfrag,epsf,amsthm]{article}

  

%\documentclass[10pt,leqno,amssymbols]{amsbook}

%\documentstyle[12pt,twoside]{amsart}
%\documentstyle[amstex,12pt,leqno]{book}
%\documentstyle[amstex,12pt,leqno]{amsart}
%\documentstyle[12pt,leqno]{amsbook}


\usepackage{amssymb}



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

% Definitions for new environments
% theorem style plain --- default
%\newtheorem{thm}{Theorem}[chapter]
\newtheorem{thm}{Theorem}%[section]
\newtheorem{mainthm}[thm]{Main Theorem}
\newtheorem{question}[thm]{Question}
\newtheorem{lem}[thm]{Lemma}
\newtheorem{cor}[thm]{Corollary}
\newtheorem{cors}[thm]{Corollaries}
\newtheorem{prop}[thm]{Proposition}
\newtheorem{crit}[thm]{Criterion}
\newtheorem{conj}[thm]{Conjecture}
\newtheorem{principle}[thm]{Principle} %!!!!!!!!!!!!!!!!!!!!!!
\newtheorem{complem}[thm]{Complement}%!!!!!!!!!!!!!!!!!!!!!!
\newtheorem{pitfall}[thm]{Pitfall}%!!!!!!!!!!!!!!!!!!!!!!
   
\theoremstyle{definition}
\newtheorem{defn}[thm]{Definition}
\newtheorem{condition}[thm]{Condition}
\newtheorem{say}[thm]{}
\newtheorem{exmp}[thm]{Example}
\newtheorem{hint}[thm]{Hint}
\newtheorem{exrc}[thm]{Exercise}
\newtheorem{prob}[thm]{Problem}
\newtheorem{const}[thm]{Construction}   %!!!!!!!!!!!!!!!!
\newtheorem{ques}[thm]{Question}    %!!!!!!!!!!!!!!!!!!!!
\newtheorem{alg}[thm]{Algorithm}
\newtheorem{rem}[thm]{Remark}          
\newtheorem{aside}[thm]{Aside}          
%\renewcommand{\theremark}{} 
\newtheorem{note}[thm]{Note}            %\renewcommand{\thenote}{} 
\newtheorem{summ}[thm]{Summary}         %\renewcommand{\thesumm}{} 
\newtheorem{ack}{Acknowledgments}       \renewcommand{\theack}{} 
\newtheorem{notation}[thm]{Notation}   
\newtheorem{warning}[thm]{Warning}  
\newtheorem{defn-thm}[thm]{Definition--Theorem}  %!!!!!!!!!!!!!!!!!!!!!!!!
\newtheorem{defn-lem}[thm]{Definition--Lemma}  %!!!!!!!!!!!!!!!!!!!!!!!!
\newtheorem{convention}[thm]{Convention}  %!!!!!!!!!!!!!!!!!!!!!!!!!!!
\newtheorem{keyidea}[thm]{Key idea} 
\newtheorem{expl}[thm]{Explanation}   
\newtheorem{meth}[thm]{Method}   


\theoremstyle{remark}
\newtheorem{claim}[thm]{Claim}  
\newtheorem{case}{Case} 
\newtheorem{subcase}{Subcase}   
\newtheorem{step}{Step}
\newtheorem{approach}{Approach}
%\newtheorem{principle}{Principle}
\newtheorem{fact}{Fact}
\newtheorem{subsay}{}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\setcounter{section}{0}
 %\documentstyle[12pt]{book}
%\setlength{\topmargin}{0in}
%\setlength{\headsep}{0in}
%\setlength{\headheight}{0in}
%\setlength{\textheight}{8.25in}
%\setlength{\parindent}{0in}
%\pagestyle{empty}


\let \cedilla =\c
\renewcommand{\c}[0]{{\mathbb C}}  
\let \crossedo =\o
\renewcommand{\o}[0]{{\mathcal O}} 
%\renewcommand{\o}[0]{{\mathcal O}} 
\newcommand{\z}[0]{{\mathbb Z}}
\newcommand{\n}[0]{{\mathbb N}}
\let \ringaccent=\r  %%% \r  shorthand for 'ring accent'
\renewcommand{\r}[0]{{\mathbb R}} 

\renewcommand{\a}[0]{{\mathbb A}} 

\newcommand{\h}[0]{{\mathbb H}}
\newcommand{\p}[0]{{\mathbb P}}
\newcommand{\f}[0]{{\mathbb F}}
\newcommand{\q}[0]{{\mathbb Q}}
\newcommand{\map}[0]{\dasharrow}
\newcommand{\qtq}[1]{\quad\mbox{#1}\quad}
\newcommand{\spec}[0]{\operatorname{Spec}}
\newcommand{\pic}[0]{\operatorname{Pic}}
\newcommand{\gal}[0]{\operatorname{Gal}}
\newcommand{\cont}[0]{\operatorname{cont}}
\newcommand{\rank}[0]{\operatorname{rank}}
\newcommand{\mult}[0]{\operatorname{mult}}
\newcommand{\discrep}[0]{\operatorname{discrep}}
\newcommand{\totaldiscrep}[0]{\operatorname{totaldiscrep}}

\newcommand{\supp}[0]{\operatorname{Supp}}    
\newcommand{\red}[0]{\operatorname{red}}    
\newcommand{\codim}[0]{\operatorname{codim}}    
\newcommand{\im}[0]{\operatorname{im}}    
\newcommand{\proj}[0]{\operatorname{Proj}}    
\newcommand{\wt}[0]{\operatorname{wt}}    
\newcommand{\socle}[0]{\operatorname{socle}}    
\newcommand{\coker}[0]{\operatorname{coker}}    
\newcommand{\ext}[0]{\operatorname{Ext}}    
\newcommand{\Hom}[0]{\operatorname{Hom}}    
\newcommand{\trace}[0]{\operatorname{Trace}}  
\newcommand{\cent}[0]{\operatorname{center}}
\newcommand{\bs}[0]{\operatorname{Bs}}
\newcommand{\aut}[0]{\operatorname{Aut}}    
\newcommand{\specan}[0]{\operatorname{Specan}}    
\newcommand{\inter}[0]{\operatorname{Int}}    
\newcommand{\sing}[0]{\operatorname{Sing}}    
\newcommand{\ex}[0]{\operatorname{Ex}}    
\newcommand{\diff}[0]{\operatorname{Diff}}  
\newcommand{\tors}[0]{\operatorname{tors}}  
\newcommand{\tor}[0]{\operatorname{Tor}}  
\newcommand{\res}[0]{\operatorname{\mathcal R}} 
\newcommand{\disc}[0]{\operatorname{Discr}} 
\newcommand{\jac}[0]{\operatorname{Jac}} 
\newcommand{\chow}[0]{\operatorname{Chow}}
\newcommand{\chr}[0]{\operatorname{char}}
\newcommand{\comp}[0]{\operatorname{Comp}}
\newcommand{\br}[0]{\operatorname{Br}}
\newcommand{\var}[0]{\operatorname{Var}}
\newcommand{\hilb}[0]{\operatorname{Hilb}}
\newcommand{\weil}[0]{\operatorname{Weil}}
\newcommand{\weilf}[0]{\operatorname{Weil_{tf}}}
\newcommand{\cl}[0]{\operatorname{Cl}}
\newcommand{\gr}[0]{\operatorname{gr}}
\newcommand{\len}[0]{\operatorname{length}}
\newcommand{\loc}[0]{\operatorname{locus}}

%\renewcommand{\conj}[0]{\operatorname{conj}} 
\newcommand{\ric}[0]{\operatorname{Ricci}}

\newcommand{\nec}[1]{\overline{NE}({#1})}

\newcommand{\rup}[1]{\lceil{#1}\rceil}
\newcommand{\rdown}[1]{\lfloor{#1}\rfloor}

\newcommand{\onto}[0]{\twoheadrightarrow}

\newcommand{\lcm}[0]{\operatorname{lcm}}
\newcommand{\wdeg}[0]{\operatorname{wdeg}}
\newcommand{\obs}[0]{\operatorname{Obstr}}
\newcommand{\ann}[0]{\operatorname{Ann}}
\newcommand{\diag}[0]{\operatorname{diag}}
\newcommand{\ord}[0]{\operatorname{ord}}
\newcommand{\mord}[0]{\operatorname{max-ord}}
\newcommand{\der}[0]{\operatorname{Der}}
\newcommand{\cosupp}[0]{\operatorname{cosupp}}
\newcommand{\pardeg}[0]{\operatorname{pardeg}}
\newcommand{\tD}[0]{{\tilde{D}}}
\newcommand{\simq}[0]{\sim_{\q}}
\newcommand{\maxmult}[0]{\operatorname{maxmult}}
\newcommand{\nklt}[0]{\operatorname{nklt}}
\newcommand{\dio}[0]{\operatorname{dioph}}
\newcommand{\ddim}[0]{\operatorname{d-dim}}
\newcommand{\pole}[0]{\operatorname{pole}}
\newcommand{\Pole}[0]{\operatorname{Pole}}
\newcommand{\lcth}[0]{\operatorname{lct}}

\newcommand{\bir}[0]{\operatorname{Bir}}    
\newcommand{\graph}[0]{\operatorname{graph}}    
\newcommand{\isom}[0]{\operatorname{Isom}}  
\def\cprime{$'$}



%\UseAMSsymbols

%\DeclareMathSymbol\subsetneq 2328
%\DeclareMathSymbol\onto 1310

%\DeclareMathSymbol\twoheadrightarrow 1310
%\DeclareMathSymbol\rtimes 226F
%\DeclareMathSymbol\ltimes 226E

\def\into{\DOTSB\lhook\joinrel\to}

%\makeindex


\begin{document}
\bibliographystyle{amsalpha}

%\title{Universal untwisting of birational maps}
\title[Birational rigidity of Fano varieties]
{Birational rigidity of Fano varieties \\
and field extensions}
\author{J\'anos Koll\'ar}

\maketitle
\today


The modern study of the birational properties of Fano varieties
started with the works of Iskovskikh; see the surveys
\cite{isk, che1} and the many references there.
A key concept that emerged in this area is birational rigidity.

 Let $X$ be a Fano variety with $\q$-factorial, terminal singularities
and Picard number 1. Roughly speaking, $X$ is called
birationally rigid if $X$ can not be written
in terms of Fano varieties in any other way. 
The precise definition is the following.

\begin{defn}\label{first.defn}
A {\it Mori fiber space} is a
projective morphism $f:X\to Y$ such that
\begin{enumerate}
\item $X$ is $\q$-factorial with terminal singularities,
\item the relative Picard number of $X/Y$ is 1, and
\item $-K_X$ is $f$-ample.
\end{enumerate}

 Let $X$ be a Fano variety with $\q$-factorial, terminal singularities
and Picard number 1 defined over a field $k$. $X$ is called
{\it birationally rigid}  if $X$ is not birational to the
total space of any Mori fiber space, save the trivial
one $f:X\to \spec k$.

For our purposes, it is better to separate this condition
into 2 parts:
\begin{enumerate}\setcounter{enumi}{3}
\item Let $X'$ be a  Fano variety with $\q$-factorial, terminal singularities
and Picard number 1 over $k$ that is birational to $X$. Then
$X'$ is isomorphic to $X$.
\item The only 
rational map $g:X\map Y$ with rationally connected general fibers 
is the constant map $X\map (\mbox{point})$.
\end{enumerate}


In characteristic 0, \cite{bchm}
implies that a map $g:X\map Y$ as in (5)
leads to a Mori fiber space $g':X'\to Y'$
with $X'$ birational to $X$ but possibly $Y'$ not birational to $Y$.
Thus in characteristic 0 the two versions are equivalent
but in positive characteristic this is not known.
\end{defn}

The aim of this note is to settle some foundational questions
about the behavior of birational rigidity in
extensions of algebraically closed fields.
In all cases when the birational rigidity of a variety has been proved,
the untwisting of birational maps 
is done by clear  geometric  constructions
that  are independent of the  algebraically closed field
of definition. Our first result  shows that, 
over  algebraically closed fields, any untwisting behaves similarly.



\begin{thm}\label{bir.thm}
 Let $k$ be an algebraically closed field
and $K\supset k$ an  algebraically closed overfield.
Let $X_k$ be  a
Fano variety  with  Picard number 1 defined over $k$.
Assume that 
\begin{enumerate}
\item either $\chr k=0$ and $X_k$ has $\q$-factorial, terminal singularities
\item or that  $X_k$ is smooth.
\end{enumerate}
 Then  $X_k$ is birationally rigid iff $X_K$ is birationally rigid.
\end{thm}

\begin{rem} I don't know how $\q$-factoriality behaves in
flat families in positive characteristic.
This is the main reason why smoothness is assumed.
For instance, a cone over an elliptic curve is
$\q$-factorial over $\bar{\f}_p$, but not
$\q$-factorial over  any other 
algebraically closed field. This example has
log canonical singularities, but I do not know
how to exclude this phenomenon for Fano varieties
with terminal singularities.
(There is also the slight problem that the definition
of terminal may not be clear in positive characteristic.)
\end{rem}

There is a very interesting problem
related to Theorem \ref{bir.thm}. 

\begin{question} \label{quest}
Let $X$ be a Fano variety over a field $k$
such that $X$ is  birationally rigid over the
algebraic closure $\bar k$. Is $X$ birationally rigid over $k$?
\end{question}


In the terminology of Cheltsov, this asks if
the notions of birational rigidity and
universal birational rigidity coincide or not.

I think that this is very unlikely but I do not have a counter example.
If $G$ denotes the Galois group $\gal(\bar k/k)$, then the
$k$-forms of $X_{\bar k}$ are
classified by $H^1(G,\aut(X_{\bar k}))$ and two such forms are
birational if they have the same image in
$H^1(G,\bir(X_{\bar k}))$. For an arbitrary variety
the map $H^1(G,\aut(X_{\bar k}))\to H^1(G,\bir(X_{\bar k}))$
is not injective. It is quite interesting that
for many birationally rigid varieties, the group
$\bir(X_{\bar k})$ is a split extension
$$
1\to \Gamma\to \bir(X_{\bar k})\to \aut(X_{\bar k})\to 1
$$
for a subgroup $\Gamma$ generated by the
``obvious'' birational self-maps. For all
such examples, $H^1(G,\aut(X_{\bar k}))\to H^1(G,\bir(X_{\bar k}))$
is an injection.
See \cite{mella, che2, shr} for several relevant examples.

Note also that the  question (\ref{quest}) is not
equivalent to the above Galois cohomology problem.
Given $X_k$, in birational geometry we are also interested in birational
equivalences $X_k\sim X'_k$
where $X'_k$ has Picard number 1 over $k$ but
higher Picard number  over $\bar k$.
\medskip

In order to study birational maps of products  of Fano varieties,
\cite{che2} introduced a variant of question (\ref{quest}).
(In the terminology of \cite{che2},
one asks for  varieties for which
$\bir(X)$ ``universally  untwists maximal centers.'')
Our second result characterizes such varieties.
For this we need to define
the dimension of $\bir(X)$.

\begin{defn} Let $X$ be a projective variety
over an algebraically closed field $k$. 
A birational map $\phi:X\map X$ can be identified with the
closure of its graph  $\graph(\phi)\subset X\times X$.
This construction realizes $\bir(X)$ as an open subscheme
of $\hilb( X\times X)$ or of $\chow( X\times X)$.
Let $\graph:\bir(X)\to \hilb( X\times X)$ denote this injection.
In general, $\graph\bigl(\bir(X)\bigr)$  is an at most
 countable union of finite type subschemes.

We can now define the {\it dimension} of  $\bir(X)$ as the
supremum of the dimensions of all irreducible subsets of
$\graph\bigl(\bir(X)\bigr)$.

This representation, however,  is not particularly unique.
Let us call a subset $Z\subset \bir(X)$ {\it constructible}
if $\graph(Z)$  is constructible as a subset of $\hilb( X\times X)$
(that is, 
a finite union of locally closed subvarieties). 
The notion of a constructible subset is independent of 
the birational model of $X$ and the constructible structure
is compatible with the group multiplication in $\bir(X)$.
The dimension of a constructible subset is also well defined.

We can  also define the  dimension of  $\bir(X)$ as the
supremum of the dimensions of all constructible subsets of
$\bir(X)$.
\end{defn}



\begin{thm}\label{thm}  Let $k$ be an algebraically closed field
of characteristic 0.
Let $X$ be a 
Fano variety  with $\q$-factorial, terminal singularities
and Picard number 1 defined over $k$.
The following are equivalent:
\begin{enumerate}
\item $X$ is  birationally rigid and $\dim \bir(X)=0$.
\item $X$ is  birationally rigid and 
$\bir\bigl(X_K\bigr)=\bir X$ for every 
overfield $K\supset k$.
\item $X_K$ is  birationally rigid and 
$\bir\bigl(X_K\bigr)=\bir X$ for every 
overfield $K\supset k$.
\end{enumerate}
If $k$ is uncountable, these are further equivalent to
\begin{enumerate}\setcounter{enumi}{3}
\item $X$ is  birationally rigid and $\bir(X)$ is countable.
\end{enumerate}
\end{thm}

\begin{say}[Comments on  pliability]
The works of Corti suggest that instead of birational rigidity,
one should consider varieties which are birational to 
only finitely many different Mori fiber spaces
up to square equivalence
\cite{co-re, co-me}. These are called {\it  varieties with
finite pliability}.

The proof of Theorem \ref{bir.thm} also shows
that this notion is independent of 
the choice of an algebraically closed field of defintion.

The proof of Theorem \ref{thm} implies that if
$X$ has finite pliability  and $\bir(X)$ is zero dimensional
then any Mori fiber space birational to
$X\times U$ is of the form $X'\times U$ where
$X'$ is a Mori fiber space birational to $X$.
\end{say}

\begin{say}[Proof of Theorem \ref{bir.thm}]\label{pf.of.bir.thm}

Assume first that $X_K$ is birationally rigid.
Let $X'_k\to Y_k$ be a Mori fiber space whose total
space is $k$-birational to $X_k$.
Then  $X'_K\to Y_K$ is a Mori fiber space whose total
space is $K$-birational to $X_K$, thus
$X'_K$ is $K$-isomorphic to $X_K$.

$\isom(X_k, X'_k)$ is a $k$-variety which has a $K$-point.
Thus it also has a $k$-point and so
$X'_k$ is $k$-isomorphic to $X_k$. Thus
$X_k$ is birationally rigid.

The converse  follows from the next, more general, result.
\end{say}

\begin{prop}\label{bir.prop}
 Let $k$ be an algebraically closed field,
$U$ a $k$-variety and $X_U\to U$ a flat family of
 Fano varieties  with terminal singularities.
If $\chr k\neq 0$ then assume in addition that
 $X_U\to U$ is smooth.
Assume that the set 
$$
R(U):=\{u\in U(k): X_u 
\mbox{ is $\q$-factorial,  birationally rigid and $\rho(X_u)=1$}\}
$$
is Zariski dense.
 Then  the geometric generic fiber  $X_K$ of   $X_U\to U$ 
is birationally rigid.
\end{prop}


Proof. It is easy to see that  the 
Picard number of $X_K$ is also 1.
Let $\pi_K:X'_K\to S_K$ be a Mori fiber space
and $\phi_K:X_K\map X'_K$ a birational map.

Possibly after replacing $U$ by a generically finite ramified cover,
we may assume that the above varieties and maps are defined over $U$.
Thus we have
$$
X_U\stackrel{\phi_U}{\map} X'_U \stackrel{\pi_U}{\to} S_U.
$$
If $S_U\to U$ has positive dimensional fibers,
then for almost all $u\in R(U)$, the fiber $X_u$ is
birational to a nontrivial Mori fiber space; a contradiction.
Thus we may assume that $S_U=U$
and the 
Picard number of $X'_K$ is also 1.

Assume for the moment that
 the Picard number of $X'_u$ is also 1 for general $u\in R(U)$.
Then $X_u\cong X'_u$ since $X_u$ is birationally rigid. 


Let us now consider the scheme
$\isom\bigl(X_U, X'_U\bigr)$ parametrizing
isomorphisms between the fibers of
$X_U\to U$ and $X'_U\to U$. Since all isomorphisms
preserve the ample anti-canonical class,
$\isom\bigl(X_U, X'_U\bigr)$ is a scheme of finite type
over $U$ (cf.\ \cite[I.1.10]{rc-book}).
 By assumption $\isom\bigl(X_U, X'_U\bigr)\to U$
has nonempty fibers over the dense subset $R(U)\subset U$.
Therefore  $\isom\bigl(X_U, X'_U\bigr)$
dominates $U$ and so the 
geometric generic fiber  $X_K$ of   $X_U\to U$ 
is isomorphic to the 
geometric generic fiber  $X^*_K$ of   $X^*_U\to U$,
as required.

We have seen that 
the Picard number of the geometric generic fiber of
$X'_U\to U$  also 1. In characteristic 0,
this implies that the Picard number of
$X'_u$ is also 1 for general $u\in U(k)$.
(If $X'_U\to U$ is smooth and $k=\c$, then $\pic(X'_u)=H^2(X'_u,\z)$
implies this. The general singular case is treated in 
 \cite[12.1.7]{ko-mo}.) 
In positive characteristic, the topological arguments of
\cite[12.1.7]{ko-mo} do not apply and I do not know
if the rank of the Picard group is a constructible function
for families of Fano varieties.

In our case, the following auxiliary argument does
the trick.

Let $Z\to X_U\times_UX'_U$ be the normalization of the closure of the graph
of $\phi_U$ with projections $p:Z\to X_U$ and
$p':Z\to X'_U$. Let $E\subset Z$  (resp.\ $E'\subset Z$)
be the exceptional divisors of  $p$ (resp.\ $p'$).
Then
$$
\begin{array}{rcl}
\rho(Z_K)&=&\rho(X_K)+\#\{\mbox{irreducible components of $E_K$}\},\qtq{and}\\
&=&\rho(X'_K)+\#\{\mbox{irreducible components of $E'_K$}\}.
\end{array}
$$
Since $\rho(X_K)=\rho(X'_K)=1$, we obtain that 
 $E_K$ and $E'_K$ have the same
number of irreducible components.
Similarly, for general  $u\in U(k)$, $Z_u$ is the graph of
a birational map from $X_u$ to $X'_u$ and
$$
\begin{array}{rcl}
\rho(X_u)-\rho(X'_u)&=&\#\{\mbox{irred.\ comps.\ of $E_u$}\}
- \#\{\mbox{irred.\ comps.\ of $E'_u$}\}\\
&=&\#\{\mbox{irred.\ comps.\ of $E_K$}\}
- \#\{\mbox{irred.\ comps.\ of $E'_K$}\}\\
&=& 0.
\end{array}
$$
Applying this to $u\in R(U)$ we obtain that
the Picard number of $X'_u$ is also 1 for general $u\in R(U)$,
as required.\qed


\begin{say}[Proof of Theorem \ref{thm}]\label{pf.of.thm}
For any projective variety $X$,  $\aut(X)$ is a scheme
with countably many irreducible components
and the identity component $\aut^0(X)$ 
is a (finite dimensional) algebraic group.
For Fano varieties,
$\aut(X)$ respects the anticanonical polarization, hence
it acts faithfully on some projective embedding.
In particular,  $\aut(X)$ is a linear algebraic group.
Thus, if $\dim\aut(X)>0$ then $X$ is ruled 
(cf. \cite{ros} or \cite[IV.1.17.5]{rc-book}). 
 A ruled variety is not birationally rigid
(except for $X=\p^1$). In particular, a birationally rigid
variety of dimension $\geq 2$ has a finite automorphism group.
$\p^1$ does not satisfy any of the conditions of
(\ref{thm}.1--4). Hence from now on we may assume that
$\aut(X)$ is finite.

 Over an uncountable algebraically closed field,
a scheme which is a countable union of finite type 
subschemes has countably many points iff
it is zero dimensional. Thus 
 (1) and (4) are equivalent.

Let us prove next that (1) and (2) are equivalent.
Let $U\subset \bir(X)$ be a positive dimensional irreducible component
with generic point $z_g$. Then $z_g$ corresponds to a
birational self-map $\phi_Z$ of $X$ defined over $k(Z)$
which is not in $\bir(X)$. Conversely, if
$\phi_K\in \bir(X_K)$ is not in $\bir(X)$ then
the closure of the corresponding point in
$\hilb(X\times X)$ gives a positive dimensional component
of $\bir(X)$.

It is clear that (3) $\Rightarrow$ (2).


Finally let us prove that (1) implies (3),



Let $X'_K$ be a Mori fiber space birational to $X_K$.
We may assume that everything is defined over 
a $k$-variety $U$.
Thus we have
$$
X_U\stackrel{\phi_U}{\map} X'_U \stackrel{\pi_U}{\to} S_U.
$$
If $S_U\to U$ has positive dimensional fibers,
then for almost all $u\in U(k)$, the fiber $X_u$ is
birational to a nontrivial Fano fiber space; a contradiction.
Thus we may assume that $S_U=U$.




Here we are not allowed to replace $U$ with an arbitrary
ramified cover, hence
we can not
assert  that $X'_u$ has Picard number 1 for general $u\in U(k)$.

There is, however, a quasi-finite Galois cover $W\to U$
 such that the whole Picard group of the
geometric generic fiber of $X'_W$ is defined over $k(W)$.
Let us run the MMP on
$X'_W\to W$ to end up with
$\tau: X'_W\map X^*_W$ and a Mori fiber space
$ X^*_W\to S^*_W$.
(This much of the MMP is known in any dimension by \cite{bchm}.)
As before we see that $S^*_W\cong W$ and
the fibers of $X^*_W\to W$ have Picard number 1.
The fibers of $X^*_W\to W$ over $k$-points are thus isomorphic to $X_k$
and so, arguing with 
$\isom\bigl(X_k\times W, X^*_W\bigr)$ as above,
we obtain that 
 $X^*_W\to W$ is an \'etale locally trivial $X$-bundle
over a dense open subset of $W$.
After a possible further covering of $W$, we may assume that 
 $X^*_W\cong X\times W$.  

Note that the Galois group $G=\gal(W/U)$ need not act trivially on
the Picard group, thus we use the ordinary MMP, not the
$G$-equivariant MMP. In particular, the $G$-action on
$X'_W$ does not give a regular $G$-action on $X^*_W$,
only a birational $G$-action.

Thus, in (\ref{pf.of.thm}.1) the maps are $G$-equivariant,
the $G$-action is trivial on the $X$-factor on the left
but on  $X^*_W\cong X\times W$ we have a so far unknown
birational $G$-action which commutes with projection to $W$. 
$$
X\times W=X_W\stackrel{\phi}{\map} X'_W\stackrel{\tau}{\map} X^*_W
\cong X\times W.
\eqno{(\ref{pf.of.thm}.1)}
$$
A birational map 
$X\times W \map X\times W$ which commutes with projection to $W$
corresponds to a rational map $W\map\bir(X)$. Since
$\bir(X)$ is zero dimensional, any such 
birational map is obtained by $\bir(X)$ acting on the $X$-factor.
Thus the $G$-action on  $X^*_W\cong X\times W$
is given by 
$$
(x,w)\mapsto \bigl(\phi(g)(x), g(w)\bigr)
$$
where $\phi:G\to \bir(X)$ is a homomorphism.

Applying this argument to $\tau\circ \phi$, we get that
$$
\bigl(\tau\circ \phi\bigr)(x,w)=\bigl(\rho(x), w\bigr)
$$
for some $\rho\in \bir(X)$. 



The $G$-equivariance  of $\tau\circ \phi$ gives
the equality
$$
\bigl(\rho(x), g(w)\bigr)=\bigl(\phi(g)\rho(x), g(w)\bigr)
\quad \forall g\in G.
$$
Thus $\phi(g)$ is the identity and so the 
$G$-action on  $X^*_W\cong X\times W$ is in fact 
trivial on the $X$-factor, hence biregular. 

We can now take the quotient by $G$ to obtain
$$
\tau/G:X'_U=X'_W/G\map X^*_W/G\cong X\times U.
$$
Note that $\tau$ does not extract any divisor,
thus $\tau/G$ also does not extract any divisor.
Since $X'_U$ has relative Picard number 1 over $U$,
we conclude that $\tau/G$ also does not contract
any divisor. By a lemma of Matsusaka and Mumford
(cf.\ \cite[5.6]{ksc}), this implies that
$\tau/G$ is an isomorphism.
\qed
\end{say}

 \begin{ack}  I thank I.\ Cheltsov
for useful e-mails and comments. 
Partial financial support  was provided by  the NSF under grant number 
DMS-0500198. 
\end{ack}

\bibliography{refs}

\vskip1cm

\noindent Princeton University, Princeton NJ 08544-1000

\begin{verbatim}kollar@math.princeton.edu\end{verbatim}



\end{document}


 starting with the easier case when
$k$ has 
 infinite transcendence degree over its prime field.
 
Pick $U$ and let  $\mathcal{M}$ be a 
linear system on $X\times U$ without fixed components.
Then $\mathcal{M}_u$, its restriction to $X_u$, is also
without fixed components for general $u\in U$.

Let $k_1\subset k$ be an  algebraically closed field
 of finite transcendence degree over its prime field
such that $X,U$ and $\mathcal{M}$ are defined over $k_1$.
Pick now $u_0\in U(k)$ such that its conjugates 
over $k_1$ are dense
in $U$. (This is possible whenever the
transcendence degree of $k/k_1$ is at least $\dim U$.)
If $\gamma(u_0)\in \bir(X_k)$ untwists $\mathcal{M}_{u_0}$ then its conjugates
untwist the conjugates of $\mathcal{M}_{u_0}$. 
We can view $\bir(X)$ as a $k_1$-scheme (of dimension $0$),
hence the automorphisms of $k/k_1$ act trivially
on it. 
Thus $\gamma(u_0)$ untwists not just $\mathcal{M}_{u_0}$
but also $\mathcal{M}_u$ for any conjugate $u$ of 
 $u_0$.  
Consider now the linear system
$\gamma(u_0)\mathcal{M}$ on $X\times U$.
Taking the image of a linear system by a birational map
does not commute with specialization in general, but
it does commute with specialization over a suitable
dense open subset $U^0\subset U$.

Since being canonical is an open condition,
we conclude that $\gamma_{u_0}$ untwists
$\mathcal{M}$ on  $X\times U^0$.

This takes care of most of the characteristic 0 fields,
including $\c$.
The proof of the general case is postponed until the end.
END ?????




For technical reasons, mainly to make the statements independent of the
minimal model program, the following is the currently accepted definition:



\begin{defn} Let $k$ be a field and $X$ a Fano variety 
 with terminal singularities
and Picard number 1 over $k$.
Let $\bir(X)$ denote the group of birational self-maps of $X$ and 
let $\Gamma\subset \bir(X)$ be  a subgroup. We say that
$\Gamma$ {\it untwists} all maximal singularities on $X$ if 
for every  linear system $\mathcal{M}$ on $X$ without fixed components,
 there is a $\gamma\in \Gamma$ and a $c\in\r$ such that
$\bigl(X, c\gamma_*\mathcal{M}\bigr)$ is canonical 
and $K_X+c\gamma_*\mathcal{M}\equiv 0$.

We say that $X$ is {\it birationally rigid} if
$\Gamma=\bir(X)$ untwists all maximal singularities.

The equivalence of the two definitions is
given essentially by  the Noether-Fano method.
(See \cite[Chap.5]{ksc} for an introduction
or \cite{cor1, cor2} for a more thorough treatment.
The proof  relies on various 
consequences of the minimal model program;
these have been  established only in characteristic 0.)

In order to study birational maps of products  of Fano varieties,
\cite{??} introduced the following variant.

We say that
$\Gamma$ {\it universally untwists} all maximal singularities if 
for every field extension $K\supset k$, 
$\Gamma$  untwists all maximal singularities on $X_K$.
It is clearly enough to check this for finitely generated
extensions.

It $k$ is algebraically closed, which is the original setting
of \cite{??}, then we can restate this in the following more geometric way.

$\Gamma$  universally untwists all maximal singularities if 
for every $k$-variety $U$ and for every 
 linear system $\mathcal{M}$ on $X\times U$ without fixed components,
 there is a $\gamma\in \Gamma$ and a $c\in\r$ such that
$\bigl(X_u, c\gamma_*\mathcal{M}_u\bigr)$ is canonical 
and $K_X+c\gamma_*\mathcal{M}_u\equiv 0$ for general $u\in U$, 
where $\gamma_*\mathcal{M}_u$ denotes the restriction of
$\gamma_*\mathcal{M}$ to a fiber $X_u:=X\times \{u\}$.
\end{defn}


\begin{say}[End of the proof of Theorem \ref{thm}]

Let ${\mathcal M}$ be a linear system on $X_U$
such that $K_{X_U}+c{\mathcal M}\equiv 0$ and
$(X_U, c{\mathcal M})$ is not canonical.

As  in the proof of (\ref{bir.prop}), we construct 
$\sigma:X^*_U\to U$.

Here we are not allowed to change $U$, hence we can not
assert  that $X^*_u$ has Picard number 1 for general $u\in U(k)$.

There is, however, a quasifinite Galois cover $W\to U$ with Galois group
$G$ such that the whole Picard group of the
geometric generic fiber of $X^*_W$ is defined over $W$.
As before, we run the MMP on
$X^*_W\to W$ to end up with
$X^*_W\map X^{**}_W\to W$ where, 
the fibers of $X^{**}_W\to W$ have Picard number 1.
As in 
the proof of (\ref{bir.prop}),
after a possible futher covering of $W$, we may assume that 
 $X^{**}_W\cong X\times W$.  


We thus have birational maps
$$
X\times W  \stackrel{\pi}{\map}
X^*_W  \stackrel{\tau}{\map} X^{**}_W \cong X\times W,
$$
where $\pi$ is $G$-equivarian but $\tau$ need not be
$G$-equivariant.
Note however that, since $\dim\bir(X)=0$, every
birational map $X\times W \map X\times W $ that comutes with projection to $W$
is obtained by $\bir(X)$ acting on the $X$-factor.
In particular, any such  birational map
commutes with any group action on $W$.
Thus the composite 
 $\tau\circ\pi:X\times W \map X\times W $ is $G$-equivariant, thus
$\pi:X^*_W \map X^{**}_W$ is also $G$-equivariant.
This means that $X^*_W \map X^{**}_W \cong X\times W$
descends to $X^*_U \map X^{**}_U \cong X\times U$.


Since $X^*_U/U$ has relative Picard number 1,
this implies that $X^*_U \map X^{**}_U$
is an isomorphism.
Thus $X^*_U \cong X\times U$.

Thus $\tau_U\circ \pi_U$ is given by the actiuon of some
$\gamma\in \bir(X)$ on the $X$-factor. Thus
$\gamma$ untwists ${\mathcal M}$.\qed

\end{say}



In particular, 
$X^*_u\cong X^{**}_u$ has Picard number 1 for general $u\in U(k)$.
Now we can conclude that $X^*_u\cong X$ for general $u\in U(k)$
and $X^*_U\to U$ is  an \'etale locally trivial
$X$-bundle over some open subset $U^0\subset U$.


Thus we have a birational map from $X\times U\to U$ to
$X^*_U\to U$ which commutes with projection to $U$.

Thus, for each $u\in U$ it is untwisted by some
$\bar \gamma\in \bir(X)/\aut(X)$.
Since $\aut(X)$ is finite, there is a
$\gamma\in \bir(X)$ which untwists on a dense set, hence
on an open set ............
\qed




 
Next we claim that $X^*_U\to U$ is an \'etale locally trivial
$X$-bundle over some open subset $U^0\subset U$.
To see this take an embedding
$X^*_U\into \p^N_U$ such that $-mK_{X^*_U/U}$ is the pull back
of $\o_{\p^N}(1)$. Thus  $X^*_U\to U$ corresponds to a rational
map $g:U\map \hilb(\p^N)$.
The subvarieties $Z\subset \p^N$ such that
$\bigl(Z,\o_Z(1)\bigr)\cong \bigl(X, \o_X(-mK_X)\bigr)$
form a subscheme $W_X$ of finite type in $\hilb(\p^N)$.
By assumption $W_X\cap g(U)$ contains all the $k$-points of $U$,
hence $W_X\cap g(U)$ is dense and open in $g(U)$.
Thus there is an open $U^0\subset U$ such that 
  $X^*_U\to U$ is an \'etale locally trivial
$X$-bundle over $U^0$. In particular,
the geometric generic fiber of
$X^*_U\to U$ is isomorphic to $X_K$ and the resulting
birational map $\gamma: X_K\leftarrow X'_k \map  X^*_K$
untwists ${\mathcal M}$.
\qed






\begin{rem} It is quite likely that (\ref{thm}.1--3) are
equivalent for any algebraically closed field $k$.
The implications  (1) $\Rightarrow$ (2) $\Leftrightarrow$ (3)
always hold.

In all known proofs of birational rigidity, the untwisting maps
depend on the linear system $\mathcal{M}$ in a very simple
algebraic way. This, however, may be an artifact of our
inability to prove birational rigidity in more complicated cases.

This is closely related to the following interesting
open problem.
\end{rem}










Let $K$ be an algebraic closure of $k(U)$.
We can then view $\mathcal{M}$ as a linear system on
$X_K$.  Note that $X$ and $\mathcal{M}$ are defined over a 
finitely generated subfield of $K$.
Since $k$ has uncountable transcendence degree over its
prime field, there is a subfield $k'\subset K$ such that
$X$ and $\mathcal{M}$ are defined over $k'$ and $k\cong k'$.
Since $X_k$ is birationally rigid, so is
$X_{k'}$. Thus there is a $\phi\in \bir\bigl(X_{k'}\big)$
which untwists $\mathcal{M}$. By base extension,
we can view $\phi$ as an element in $\bir(X_K)$. 
From (3) we conclude that  $\phi\in \bir(X)$, giving (1).

Taking the closure of the graph
does not, in general, commute with specialization, but it
does commute over an open dense set. Thus the closure of
of the graph
$$
\Gamma(\Phi)\subset (X\times U)\times_U (X\times U)\cong X\times X\times U
$$
is the graph of an automorphism for general
$u\in U$. 

As we noted, the  automorphism group of $X$ is finite.
Hence, there is an automorphism $\sigma\in \aut(X)$ such that
$$
\sigma \circ \gamma\circ \Phi^{-1}_*:X\times U\map X\times U
$$

<- direction is OK.
If Bir(X) untwists and is countable,
then Bir(X) universally untwist.

-> direction is not clear to me.
COnsider example.
Let X->Q be a generic double cover of smooth
quadric such that X is a Fano 3-fold of degree 4.
Then Bir(X) is a semidirect product of Aut=Z_2
and free product of uncountably many birational
involutions induced by lines C on X (-CK=1).
Then X is universally birationally ridid (because there are
no relations between involutions induced by lines).
It seems to me that Bir(X) must univerally untwist.
This is equivalent to the following question:
Let F be a subvareity
in X\times U such that the general fiber of
the projection F->U is a line in X.
Is it true that F=C\times U for some line C on X?

What do you think?

Ivan

Quoting Janos Kollar <kollar@Math.Princeton.EDU>:

> Ivan
>
> it seems to me that, except for P^1
>
> Bir(X) universally  untwists <-> Bir(X) untwists and is countable
>
> You need <-.
>
> Proof. Since Bir(X) untwists, we know that
> Bir(X_K) unwists where K is the alg closure of the function field C(U).
>
> So the question is: Bir(X)=Bir(X_K)?
>
> The graphs of Bir are parametrized by a subsche of Hil(X\times X)
> which is a countable union of finite type schemes.
> So K-points=C-points iff it is zero dimensional,
> iff Bir(X) is countable. (Here I use that C is not countable.)
>
> Other direction. If Bir(X)\neq Bir(X_K), take any free linear system
> and apply some g\in Bir(X_K)-Bir(X). This needs to be untwisted.
> So if done by g'\in Bir(X) then (maybe this is not quite right)
> g'g\in Aut(X). So Aut(X) is positive dimensinal, so X is ruled.
> Except for P^1, this implies that X is very nonrigid.
>
>
>
> -- 
> János
>
> Department of Mathematics, Fine Hall, Washington Road
> Princeton University,  Princeton, NJ 08544-1000  USA
> e-mail:  kollar@math.princeton.edu



Dear J\'anos,

Thank you!

Yes, I see. I got the point.
The proof of -> is right.
May I put add remark somewhere about this?

So in my example if X is general and U is say
rationally connected then Bir(X\times C(U))=Bir(X),
because otherwise we get non-constant map U-->Z,
where Z is curve parametrizing lines on X, which is
not rational.

Now the question is it possible to get birationally rigid Fano
which is not univerally birationally rigid.
This question is very similar to one considered by Max Mella in his
paper on factorial quartic 3-folds.
He proves that factorial nodal quartic 3-fold is bir.rigid over
any field not just over C (and I think that you suggested this
problem?).
However this part of Max's paper has mistake.
I wrote to him and he knows about it.
Namely he assumed that 2 singular points can not be non-canonical
centers simultaneously, but they CAN (if there is a line through them).
This problem is related to the problem of finding relations
between birationa involutions (generators) over C.

